Author's response to reviews

Title: Eating patterns of Turkish adolescents: a cross-sectional survey

Authors:

mehmet akman (makman4@gmail.com)
hulya akan (hakan@yeditepe.edu.tr)
guldal izbirak (gizbirak@yeditepe.edu.tr)
özlem tanrio ver (otanrio ver@yeditepe.edu.tr)
sirma mine tilev (stilev@yeditepe.edu.tr)
anil yildiz (ayildiz@yeditepe.edu.tr)
simge tektas (stektas@yeditepe.edu.tr)
ayca vitrinel (avitrinel@yeditepe.edu.tr)
osman hayran (ohayran@yeditepe.edu.tr)

Version: 5 Date: 15 December 2010

Author's response to reviews:

Dear Editors,

We are delighted to hear that our manuscript is accepted for publication. We have done the required format changes and all the major and minor revisions requested.

since some Turkish letters are different than English ones, names of the authors are written with Turkish letters in the manuscripts's title page whereas they were written with using only English letters in the submission system. If possible Turkish version is prefered in the published article.

Best regards

mehmet akman